Author's response to reviews

Title: Down syndrome with microgranular variant of acute promyelocytic leukemia in a child - A case report

Authors:

Deepali Jain (deepalijain76@gmail.com)
Tejindar Singh (tsinghmamc@yahoo.co.in)
Prema Arora (parora@rediffmail.com)

Version: 6 Date: 30 October 2007

Author's response to reviews: see over
To,
The Editor in Chief,
Journal of Medical Case Reports

SUB: Resubmission of article: **MS: 8257698421577082** after revision

Respected Sir,

Thank you for your kind response to the article, titled as "**Down syndrome with microgranular variant of acute promyelocytic leukemia in a child: a case report.**"

According to your suggestions grammar and spelling changes are marked in red.

I would be pleased if you can please consider it for publication in your esteemed journal.

Thanking you
Yours sincerely,
Dr. Deepali Jain, M.D.

Department of Pathology
Maulana azad medical college
New Delhi, India 110002

Email: deepalipgi@gmail.com
Phone: 91-98688-95112